Author’s response to reviews

Title: Associations between Occupational Stress, Burnout and Well-being among Manufacturing Workers: Mediating Roles of Psychological capital and Self-esteem

Authors:

Ziyue Wang (yuewzy@foxmail.com)
Hongbo Liu (hbliu@cmu.edu.cn)
Haijian Yu (YHJ19891221@yahoo.cn)
Yanwen Wu (wuyanwen_cmu@163.com)
Shuai Chang (chang9054@163.com)
Lie Wang (liewang@cmu.edu.cn)

Version: 4 Date: 03 Nov 2017

Author’s response to reviews:

Dear Editor,

We appreciate the editors very much for the suggestions on our manuscript entitled “Associations between Occupational Stress, Burnout and Well-being among Manufacturing Workers: Mediating Roles of Psychological capital and Self-esteem” (BPSY-D-17-00203R3). We revised our manuscript according to comments. We would like to express our great appreciation to you and reviewers for comments on our paper.

Looking forward to hearing from you.

Response editors

1. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

Response the comments: p14,line387-388.

2. Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos). Therefore,
please state "Not applicable" in this section as no personal information is provided in your manuscript.

Response the comments: p14, line395.

3. Please remove any figure titles and captions from the figure files as these files should contain the image graphics only. Upon doing this, please place the titles/captions at the end of the main manuscript after the References section under a newly created “Figure legend” heading.

Response the comments: All figures titles and captions has been added after the references section.p18.line412-413.

4. Please upload English language versions of all questionnaires developed specifically for use in this study as supplementary files. If the questionnaires are not yours or are already published by you, please supply references and/or links to them.

Response the comments: links/references of questionnaires have been as supplementary files.

5. Please add a “Supplementary files” section where you list the following information about your questionnaire, if included for publication:

Response the comments: p19.line415-416

6. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

Reponse the comments: Thankyou, we have done this.

Yours sincerely,

Lie Wang.